Design development and implementation of an irradiation station at the neutron time-of-flight facility at CERN

M. Ferrari,^{1, *} D. Senajova,^{1, 2} O. Aberle,¹ Y. Q. Aguiar,¹ D. Baillard,¹ M. Barbagallo,^{1, †}

A.-P. Bernardes,¹ L. Buonocore,¹ M. Cecchetto,¹ V. Clerc,¹ M. Di Castro,¹ R. Garcia Alia,¹

S. Girod,¹ J.-L. Grenard,¹ K. Kershaw,¹ G. Lerner,¹ M. Maeder,^{1,‡} A. Makovec,¹

A. Mengoni,^{1,3} M. Perez Ornedo,¹ F. Pozzi,¹ C. V. Almagro,^{1,4} and M. Calviani^{1,§}

(for the n_TOF Collaboration)

¹European Organization for Nuclear Research, CERN,

Esplanade des Particules 1, 1211 Geneve 23, Switzerland

²Imperial College London, SW7 2AZ, London, United Kingdom

³ENEA, Italian National Agency for New Technologies,

Energy and Sustainable Economic Development, Italy

⁴ Jaume I University of Castellon, 12006 Castellon de la Plana, Spain

(Dated: June 22, 2022)

A new parasitic, mixed-field, neutron-dominated irradiation station has been recently commissioned at the European Laboratory for Particle Physics (CERN). The station is installed within the Neutron Time-Of-Flight (n_TOF) facility, taking advantage of the secondary radiation produced by the neutron spallation target, with neutrons ranging from 0.025 eV to several hundreds of MeV.

The new station allows radiation damage studies to be performed in irradiation conditions that are closer to the ones encountered during the operation of particle accelerators; the irradiation tests carried out in the station will be complementary to the standard tests on materials, usually performed with gamma sources.

Samples will be exposed to neutron-dominated doses in the MGy range per year, with minimal impact on the n₋TOF facility operation. The station has twenty-four irradiation positions, each hosting up to 100 cm^3 of sample material.

In view of its proximity to the n_TOF target, inside protective shielding, the irradiation station and its operating procedures have been carefully developed taking into account the safety of personnel and to avoid any unwanted impact on the operation of the n_TOF facility and experiments. Due to the residual radioactivity of the whole area around the n_TOF target and of the irradiated samples, access to the irradiation station is forbidden to human operators even when the n_TOF facility is not in operation. Robots are used for the remote installation and retrieval of the samples, and other optimizations of the handling procedures were developed in compliance with radiation protection regulations and the aim of minimizing doses to personnel.

The sample containers were designed to be radiation tolerant, compatible with remote handling and subject to detailed risk analysis and testing during their development. The whole life cycle of the irradiated materials, including their post-irradiation examinations and final disposal was considered and optimized.

I. INTRODUCTION

Despite their known sensitivity to radiation [1, 2], non-metallic materials are, of necessity, extensively used in high-radiation areas of accelerators and high-power physics facilities [3, 4]. Commercial polymeric components such as lubricating oils, greases, elastomeric Orings, insulators and optical fibers are used in crucial equipment in areas where intense levels of radiation are encountered at research laboratories worldwide. Examples of large infrastructures where the selection of radiation tolerant materials is critical for successful operation include: the high intensity proton accelerator facility at the Japan Proton Accelerator Research Complex (J-PARC, Japan), the accelerator complex at Fermilab (USA), the Spallation Neutron Source at the Oak Ridge National Laboratory (ORNL, USA), the ITER collaboration (France), the European Spallation Source (ESS, Sweden) as well as the accelerator complex at the European Laboratory of Particle Physics (CERN, Switzerland). Additionally, various other types of more compact high-power target facilities currently operating or under development, such as ISOLDE (Isotope Separation On-Line) facilities [5, 6] and MEDICIS (Medical Isotopes Collected from ISOLDE) [7], share comparable challenges in the selection of materials to be used for their construction and operation.

Radiation-induced degradation of specific components can limit the lifetime of accelerator equipment such as Beam Intercepting Devices (BIDs) [8–10], including complex high-power target assemblies, necessitating their

^{*} matteo.ferrari.2@cern.ch

[†] Currently at Transmutex, Geneva, Switzerland.

[‡] Currently at the University of Applied Sciences Offenburg, Department of Electrical Engineering, Medical Engineering and Computer Science, Badstraße 24, 77652 Offenburg, Germany.

[§] marco.calviani@cern.ch

earlier replacement [11, 12]. In the above-mentioned devices, commercial materials are typically exposed to mixed radiation fields, absorbing physical doses ranging up to several MGy during their lifetime [8, 10]. Many types of polymeric materials are reported to fail at comparable doses [1].

Historically, accelerated irradiation tests of commercial materials have been performed for the development of nuclear, aerospace, fusion and accelerator technologies [3, 13–17]. At CERN, extensive tests of materials and components to be used in accelerators have been performed between the 1960's and the early 2000's. The results are reported in a series of documents known as Yellow Reports; examples are given in References [18– 20]. Most of the results have been collected using gamma sources, under the assumption that equal doses roughly induce equal damage, regardless of the radiation type [2, 20].

However, recent studies [21–23] question this hypothesis and report a general lack of up-to-date scientific knowledge in this field [24], highlighting the need for new data to be collected under different irradiation conditions. Due to the continuous upgrading and development of accelerators and to the increased intensity of the produced radiation, new irradiation studies are necessary to correspond with the more demanding requirements on materials for use in accelerators and to minimize radiation-induced failures [25, 26].

For example, irradiation tests on materials using gamma sources are generally accelerated so that total doses comparable to the ones absorbed in years or decades of operation can be delivered in much shorter times in testing conditions. Data collected in mixed fields and longer irradiation times are currently very scarce, but they would better reproduce the radiation effects occurring in real life conditions and they would contribute filling a general lack of knowledge.

In recent times, in-house research facilities at CERN have been occasionally used for material irradiation. In particular the IRRAD facility [27] and the CHARM (CERN High energy Accelerator Mixed field) facility [28] are used for proton and mixed field irradiation, respectively. Despite their strategic importance for the testing of electronics and detectors, they have limitations concerning the irradiation of materials. In both IRRAD and CHARM, the available particle fluences do not allow uniform doses in the MGy range to be delivered to macroscopic material samples. For this reason, new research facilities better tailored to the specific requirements of material irradiation are needed.

In the present paper, a new irradiation station recently built at CERN for various applications including material studies is presented. The station, whose installation was finalised in June 2021, allows multiple material samples to be irradiated at different dose rates in a neutron-dominated environment, over exposure times ranging from months to several years. The available radiation fields are comparable to the ones actually present in high radiation areas at CERN, in terms of both dose rate and radiation spectra. The new irradiation station at n_TOF will therefore meet the twin requirements of producing data that will facilitate the selection of materials for high-radiation areas at CERN and that will contribute to increasing the general scientific understanding in this domain.

To the best of the authors' knowledge, irradiation of liquid or semi-solid samples such as lubricants in mixed fields up to MGy levels, generating sample activation, and requiring containers being compatible with robot handling is not common, at least in a published form. The authors are not aware of any available standard or scientific reference on irradiation set-ups complying with a similar combination of technical constraints, and believe that this is the first time that such a methodology is being systematically addressed.

Section II describes the area at the n_TOF facility where the irradiation station has been constructed, hereby called "NEAR", along with other areas dedicated to different irradiation and activation experiments. Section III describes in details the technical requirements and the design aspects of the irradiation station and of the containers used for both solid and liquid material samples. Section IV describes the handling procedures necessary for the installation of the irradiation station and for the installation and retrieval of the samples, that must be performed by telemanipulation due to the high activation of the target area. Section V describes the Monte Carlo calculations performed to assess the level and homogeneity of the neutron and gamma dose absorbed by the samples in different irradiation positions. Section VI describes the radiation protection aspects of the operating procedures of the irradiation station, reporting the residual dose rate in the NEAR area and the calculated activation of samples and containers, that require special handling and disposal as radioactive materials. Section VII describes the first irradiation of material samples and the first dose measurements carried out between July and November 2021, allowing the whole procedure to be experimentally verified. Finally, conclusions are presented in Section VIII.

II. THE N_TOF FACILITY AND ITS NEAR AREA

A. n_TOF facility: introduction and recent upgrades

CERN is equipped with a top-class, high-brightness, neutron spallation source dedicated to high-resolution neutron time of flight experiments: the n_TOF Facility [29]. The facility was built in 2000 [30] and has largely evolved over the last 20 years. In particular, during CERN's Long Shutdown 2 (2019-2021), a major upgrade was implemented to guarantee reliable operation of n_TOF for the following years, improving the physics reach of the infrastructure.

Recent upgrades include the construction of a thirdgeneration spallation target [9, 31], the consolidation of the neutron collimation systems, the complete overhaul of the target pit shielding as well as the construction of a new irradiation station in the NEAR area, close to the neutron spallation target.

B. The NEAR area

Figure 1 shows a schematic overview of the n_TOF experimental areas. The area close to the new n_TOF neutron spallation target is referred to as NEAR. The New Target Mobile Shielding (simply the shielding in the paper), separates the NEAR Irradiation area (i-NEAR), which includes the target and where the irradiation station is installed from the NEAR Activation area (a-NEAR), which includes a neutron collimator and a volume outside of the target shielding.

Figure 2 shows NEAR before the installation of the new irradiation station for materials. As described in Section III, the irradiation station, located in i-NEAR, consists of two shelves (referred to as top and bottom shelf) installed on the side of the existing original target pool where the new third-generation n_TOF target [9] is located. The irradiation station installed in i-NEAR is shown in Figure 5.

The target pool (see Figure 2 b and c) is constituted by an aluminium alloy (EN AW-6082) structure which was containing the first generation n_TOF spallation target [32]; the uncladded target was immersed in demineralized water in order to remove the heat generated by the beam interaction with the Pb as well as to moderate neutrons to be employed in the first experimental area. In later generation targets, the pool was left in place in order to act as an external containment in case of water leaks from the cooling or from the moderator circuits.

In addition to the mentioned irradiation station, this highly versatile area includes additional irradiation positions in a-NEAR and channels passing through the shielding, as briefly described in II E.

C. The mobile shielding

The mobile shielding consists of two layers covering the whole length of the target chamber and one partial layer directly in front of the target. The first (inner) layer consists of 400-mm-thick steel (GG20), the second (outer) layer is 800-mm-thick concrete and the third partial layer is 200-mm-thick marble. The shielding is divided in three parts, as presented in Fig. 3. Part A is the section that is regularly moved to access i-NEAR, while part B is generally never operated, and it is there to guarantee full access to the spallation target in case of need. Part C - only constituted by 1200-mm-thick concrete - is to be moved only for the final dismantling of the facility. The

mobile shielding is mounted on top of heavy-duty roller blocks that run on steel rails fixed to the floor. Part A of the mobile shielding is opened and closed using a manually operated winch. The rails allowing the shielding movement are visible in Figure 2, in the open configuration. The rails are 100 mm wide and 50 mm high, and are an obstacle for the access to the irradiation station by robots, as described in Section IV.

The shielding can be opened during machine stops such as Year-End Technical Stops (YETS), as described in Section VI, to allow access to i-NEAR. Figure 2 shows the shielding in both closed and open positions. With the shielding in the open position, the following tasks can be carried out: access to the target pit, remote inspections of the spallation target, of the moderator and cooling circuits serving the spallation target systems. Moreover, the shielding is designed for manual operation from distance to reduce dose to personnel during interventions and final dismantling.

D. The collimation system

The shielding is designed to permit the installation of a dedicated and flexible collimation system in the proximity of the third generation target [9], in a cylindrical hole of about 280 mm in diameter (see Figure 2). The collimator (see Figure 4) consists of two cylindrical layers with a truncated cone-shaped hole in the middle, aiming at giving the desired shape to the radiation coming from i-NEAR and entering a-NEAR. The first (inner) layer consists of a 500-mm-long stainless steel component, the second (outer) layer is made of 300 mm of 5% borated polyethylene. The configuration has been studied to provide a shaped neutron beam for n_TOF experiments [33]. When not in use, the collimator can be removed and replaced with a concrete (or equivalent) cylindrical structure, to ensure the necessary shielding and to reduce the dose to personnel.

E. Rabbit tubes and other possible irradiation positions

Apart from the irradiation positions provided by the shelves, other positions are currently being designed and developed to optimise the radiation fields available in both NEAR areas. For example, several channels passing through the shielding, referred to as rabbit tubes, have been reserved to irradiate samples in i-NEAR and allow their retrieval without the opening of the shielding (see Figure 2). Additionally, specific positions in a-NEAR can be used to explore radiation damage to electronics.

An area dedicated to electronics testing is in fact available on the wall opposite to the neutron collimator, for which a patch panel is available, enabling connection to the control room and therefore the possibility of performing active tests on electronics. An assessment using com-



FIG. 1. Overview of the NEAR area within n_TOF. The upper part of the Figure shows a general scheme of the n_TOF facility. The proton beamline, the position of the neutron spallation target and the NEAR area (in the brown square) are shown. The lower part of the Figure shows different views of the NEAR area. The position of the irradiation station is indicated by the red rectangle. The mobile shielding, whose position is indicated by the arrow, separates the internal i-NEAR (the pink volume), where the irradiation station is located, from the outer a-NEAR (the green volume).



FIG. 2. a) The New Target Mobile Shielding in the closed position. b) The shielding in the open position, revealing i-NEAR and the target pool, close to the n_TOF target as it was before the installation of the irradiation station. c) 3D model of NEAR, with the shielding open, showing the position of the target pool, of the future collimator (in yellow), of the openings for rabbit tubes and of the rails allowing the shielding movement.



FIG. 3. New configuration of the mobile shielding in open position, with part A, B and C, as explained in the text. The different parts and the materials used are indicated. The outermost layer of marble is only present for part A and for part B (for the latter case, only in the area facing the spallation target).



FIG. 4. a) Steel part of the collimator photographed from a-NEAR before installation of the borated polyethylene part. The hole in the marble part of the shielding is visible at the front of the image. b) Borated polyethylene part of the collimator installed in the shielding. c) Section view of the NEAR area showing the target, shielding (steel in red, concrete in grey and marble in light grey) and access passage. The two parts of the collimator are visible inside the shielding: the steel component (in yellow) and the borated polyethlene component (in white).

mercial electronics previously tested in different spallation neutron facilities was performed, confirming that the neutron flux and spectra (see Sections V C and V D) are suitable for Single Event Effects testing for atmospheric and high-energy accelerator applications [34].

The present paper concentrates on the description of the design and construction of the shelf positions of the irradiation station in i-NEAR. Further details of the other irradiation positions will be presented in separate publications [33].

III. IRRADIATION STATION FOR MATERIALS: THE SHELF POSITIONS

A. Opportunities offered by the new irradiation station

The aim of the irradiation station is to profit from the intense secondary mixed radiation available at NEAR to study radiation damage of materials. Long irradiation exposures ranging from months to years will allow total absorbed doses in the MGy range to be delivered to organic and inorganic materials. As described in detail in Section V, most of the dose is expected to be delivered by neutrons and to a lesser extent by photons.

With the NEAR irradiation station, CERN is now equipped with twenty-four irradiation positions, where unprecedented radiation damage data can be collected. In fact, in contrast to standard gamma sources, experimental targets generate unique combinations and spectra of mixed radiation fields, which are hard to replicate in testing environments.

In the irradiation station, a selection of materials used at CERN in high radiation areas will be tested, to further understand the radiation damage mechanisms and to improve their qualification in terms of radiation resistance. The irradiation conditions available at NEAR are much closer to the operational ones when compared to gamma radiation alone. Examples of the irradiation studies to be performed at n_TOF NEAR are provided in Section VII.

B. Technical challenges and constraints

The design and operation of the new irradiation station has to comply with several technical requirements, discussed as follows.

The existing radiation fields available at n_TOF are produced for the purpose of studying neutron-nucleus interactions, with a broad range of applications [35]. To minimize any perturbation to the physics performance of the facility, the designs and positions of the new irradiation structures have been extensively crosschecked with Monte Carlo simulations.

The use of the irradiation station depends on the $n_{-}TOF$ facility schedule. The access to the irradiation

station will be typically limited to Year-End Technical Stops only, with minimal impact on the operation of the facility and on other physics experiments realized in the same experimental area.

The irradiation station can only be accessed when the shielding is open. As further discussed in Section VI, at around 40 cm from the spallation target residual dose rates typically range between tens and hundreds of mSv/h, after cool-down times ranging between days and weeks. These values are too high for human operators, so access to the irradiation station is generally forbidden.

Accordingly, the irradiation station was designed to be fully compatible with remotely handled installation. Both the installation of the station structure and the installation and retrieval of the samples are carried out with the use of telemanipulation systems, as described in Section IV.

The final disposal of the whole irradiation station is taken into account in the design phase as well. To limit the residual activation, aluminium is preferred to stainless steel. The total amount of radioactive waste is limited by opting for a modular structure compatible with potential future modifications.

The installation of the irradiation station with a telemanipulation system required some unused components still present in the area, such as old cables and cooling pipes, to be removed, and the i-NEAR to be cleaned afterwards. The pipes to be removed were part of the cooling system used for the previous design of the target; the cooling system for the third generation target is completely different and has a new set of cooling pipes. Their removal eased the access to the irradiation station for the robots, as detailed in Section IV.

C. Design of the irradiation station

The new irradiation station is attached to the original target pool by means of hooks at the top of the station frame that engage with the upper edge of the pool wall, which is no longer used for target cooling, as shown in Figure 5. In the volume of the shelves, radiation fluences have been calculated to be sufficiently intense and homogeneous for irradiation of materials, based on FLUKA [36–38] Monte Carlo simulations described in Section V.

The irradiation station includes two shelves (top and bottom shelf, respectively), as shown in Figure 5, and an additional support able to host a dedicated moderator system for a-NEAR, which is outside of the scope of this paper [33]. Each shelf can accommodate up to twelve samples, organized and handled in groups of six to facilitate installation and retrieval, when handled remotely as described in Section IV. Up to twenty-four samples can be irradiated at the same time allowing a combination of different exposure times and dose rates, as described in Section V.

The whole structure is constructed of aluminium



FIG. 5. Picture of the irradiation station, filled with samples, installed on the original target pool at n_TOF NEAR. The top and the bottom shelves are visible. Four intermediate aluminum supports (two per shelf) with their handle for remote handling, accommodate up to six containers each. (CERN-PHOTO-202107-085-11) [39].

EN AW-6082 T6, has dimensions 61x107.5x57.9 cm and weighs 15.5 kg. Fasteners are made of A4 stainless steel and there are small triangular guides for the installation of the supporting plates made of brass. The structure can safely support a full load of samples and a moderator material block weighing up to 40 kg, to be possibly installed at a later stage. The weight is compatible with the existing original target pool structure, without excessive deformations.

Eight locations hosting Radio-Photo-Luminescence (RPL) dosimeters [40] (see Figure 6) are attached to the shelf structure (four per shelf), allowing the total dose to be mapped during each use of the irradiation station. Readout measurements of the dosimeters irradiated during the first use of the irradiation station are reported in Section VIID. The RPL dosimeters are small cylinders made of phosphate glass doped with silver (Ag) ions to improve their luminescence proprieties. When they are irradiated, RPL centers are created within the crystal lattice proportionally to the absorbed dose. The dosimeters were calibrated in a ⁶⁰Co gamma-field in the Risø HDRL Facility, Denmark [41] and their measurement range extends up to a maximum of some MGy of total absorbed dose. The readout measurement is performed at CERN using a custom readout system designed specifically to address the applicability of such dosimeters in high radiation levels [42].

The access to the irradiation station for sample installation and retrieval requires the shielding to be open,



FIG. 6. Position of the dosimeters in the irradiation station, along with their labels.

and for this reason is only possible during scheduled long technical stops of the n₋TOF facility, such as Year-End Technical Stop.

D. Sample containers: design and use

Sample life-cycle analysis was used to identify requirements for the design of the containers. The main lifecycle stages to be considered are: container production, robotic or remote installation, retrieval and handling, sample irradiation, post-irradiation analysis and final disposal or decontamination.

The main required features and technical constraints are listed as follows:

- Compatibility with a remote handling. Different remote handling techniques are effectively and routinely used to safely deal with activated samples in several facilities, such as research nuclear reactors [43]. Hydraulic tube facilities to move shuttle capsules containing the samples are examples of these techniques. In this specific case, the setup has to be compatible with robot installation; this limits the containers in weight, size and shape. More details are provided in Section IV.
- Compliance with safety standards; containers must be leak-tight and fall-proof, to reduce contamination risks in case of accidental sample drop during installation or retrieval. A double layer of container is necessary, to confine possible contamination to the disposable inner container only. This allows

outer containers to be reused after decontamination, following a cool-down phase.

- Radiation tolerance; elastomeric seals should generally be avoided, as even the most radiation tolerant ones are expected to degrade at doses in the MGy range [20, 44] (see Section V). A full-metal container body and graphite seals are therefore used to ensure leak-tightness.
- Compatibility with possible pressurisation due to radiation-induced gas production; A stainless steel miniature valve [45] with dimensions Ø8.7×10 mm, a weight of 2.6 g and with an opening pressure of 0.1 bar, is fitted to the container lid, allowing a progressive gas release during irradiation, without compromising their liquid leak-tightness.
- Optimization for final disposal; aluminium is selected as the main construction material for the containers, to minimize the residual activation originating from neutron irradiation. This is a standard choice for sample irradiation in research reactors [46]. The containers are kept as small and light as possible, to reduce the radioactive waste, in agreement with general radiation protection (RP) principles.
- Compliance with radiation safety standards; the containers are compatible with handling using custom-made, long-handled 'reachers' and their opening and handling must be easy and quick. This to minimize the radiation dose to the operator, in accordance with RP guidelines. Further details are provided in Section VI.

In the final design of the containers, shown in Figure 7, all the above-mentioned aspects have been implemented.

The inner container is a commercially available cylindrical aluminium container with an external diameter of 47 mm and length of 90 mm. A seal of polymeric material is normally glued to the container lid, to ensure leak-tightness. To make the whole configuration radiation tolerant, this polymeric seal has been removed. The lid of the inner container has been pierced to allow the escape of any gas potentially produced during irradiation into the larger outer container. The total inner volume of the container is about 140 cm³. Samples with a volume up to 100 cm³ can be safely hosted by this container.

The outer container is a custom-made cylindrical aluminium container with an external diameter of 70 mm and length of 142 mm. The graphite seal ensures leaktightness, while the small valve installed in the lid allows any gas produced inside to escape once the over-pressure reaches the valve opening pressure, without compromising the overall container leak-tightness - for example if the container is dropped during remote handling. The upper part of the lid and of the container body have been shaped to be easily handled, installed and removed from the the irradiation station, and opened.



FIG. 7. Sample container design. Left: picture of the aluminium inner container and its perforated lid. Right: picture of the aluminium outer container configuration with its components. From top to bottom: the steel valve, the aluminium lid and the graphite seal.

The inner container can be inserted and extracted from the outer container using custom-designed long-handled reachers (as shown in Figure 8). The reachers enable the extraction of the inner container without spilling non solid samples and additionally allow an overall distance of 40 cm to be maintained between the operator and the containers. This might be necessary during the postirradiation handling, depending on the residual activation of the containers and samples.



FIG. 8. Extraction of an inner container from an outer container under fume hood, by use of the reachers.

The samples are installed in the irradiation station after first installing them in a intermediate support. There are a total of four intermediate supports accommodating up to six containers each. An intermediate support consists of an aluminium plate with six circular holes, allowing the sample containers to hang from the plate. Each intermediate support has a handle for remote handling, as described in Section IV C.

The intermediate support and its containers have been designed for ease of installation and removal from the irradiation station shelves. The containers have been designed and tested to ensure that samples will not spill out if they are dropped during handling.

IV. INSTALLATION AND HANDLING VIA TELEMANIPULATION SYSTEMS

Manual insertion and retrieval of the samples from the shelves is not possible due to the expected high residual activity of the n_TOF NEAR area, as described in Section VIA. Accordingly, the installation of the irradiation station and the installation and retrieval of the samples are performed by a combination of robots [47–49] operating in front of the target area at NEAR, as shown in Figure 9.

A. Preparation of i-NEAR

i-NEAR had to be prepared to host the irradiation station. The robots used needed a combination of strength, precision and specific abilities to remove the old cabling surrounding the target area, as well as the old cooling pipes installed in the pit. The tasks and the robots used to perform them are listed below:

- Remove old unused cables from the area. To cut cables, dedicated shears are installed on CERN-Bot 2.0 [49, 50]. Teodor [47], a remote operated vehicle manufactured by Telerob company [51] and CERNBot 2.0 were used in combination to respectively hold and cut the cables (See Figure 9-A).
- Vacuum cleaning of dirt particles from the pit retention vessel (located below the target pool) and the shielding floor area performed using Teodor.
- Dismantling of the cooling pipes used for a previous target design and no longer needed. It was necessary to loosen the clamps attaching the pipes to the pit. This action required a combined used of different robots: whereas Telemax [48] (a robot manufactured by Telerob company) used an angular screwdriver to reach the clamps, Teodor held the pipes to prevent them from falling on Telemax, as well as to avoid creating undesired debris.

After preparation of the area, the irradiation station was installed, as described in the following Section.

B. Installation of the shelves

All the robots entering i-NEAR have to be able to pass over the floor rails allowing the opening and closing of the shielding.

The structure of the shelves has been designed to be compatible with robots: their center of gravity and the area's dimensional constraints were carefully taken into account to meet this requirement. The structure was installed directly on the original target pool. The use of the Teodor robot, having a range of payload up to 100 kg depending on the arm stretch, was required for their installation, as shown in Figure 9-B.

The handling of the shelves requires a combination of high precision, and the ability to carry and handle a load of 15.5 kg, corresponding to the total shelf weight.



FIG. 9. Three of the main tasks carried out by robots: A) CERNBot2.0 with double arm configuration cutting cables; B) Teodor installing the shelf structure on the n_TOF's spalation original target pool; C) Telemax robot installing an intermediate support with 6 sample containers in the NEAR irradiation station (CERN-PHOTO-202107-085-7) [39].

C. Installation of the samples

As shown in Figure 10, the samples are remotely handled in groups of six by use of an intermediate support. The total weight of the intermediate support with the containers and the samples is about 3.6 kg, making it compatible with the use of Telemax, whose lifting capacity ranges up to 20 kg.

The installation of the samples in the shelves is a delicate and challenging step of the handling procedures. In fact, the samples are unavoidably shaken by the movement of the wheel tracks and by the difficult passage of the robot over the shielding rails, necessary to access the target pit. As some of the samples might be semi-solid or fluid (see Section VII), they should be kept as much as possible in a steady vertical position. In addition, the narrow available space in i-NEAR adds complication and limitations to the robot movement and degrees of freedom. As shown in Figure 9-C and in Figure 10, Telemax



FIG. 10. Telemax holding a support structure containing six samples of materials for installation (CERN-PHOTO-202107-085-1) [39].

is chosen to complete the installation and retrieval of the intermediate supports containing the samples of materials to be irradiated, due to its higher dexterity and accuracy.

D. Recovery scenarios

Given the technical difficulties related to the installation and retrieval of the samples, a complete risk analysis was performed combined with extensive tests of the most likely failure scenarios. A mock-up of the system was used to simulate the potential failure scenarios and training runs were made to prepare for the installation. As a result of this work, the designs of some parts (mainly alignment guides and holding components) were modified to be more robot-friendly and making it is easier for the operator to approach the shelf with a new group of samples and to precisely place each sample group carefully in the irradiation station.

This was necessary before the go-ahead was given to start work in the active area. Figure 11 shows recovery trials after simulating a dropped intermediate support and its containers.

In conclusion, the set of available robots, combined with the specific training of the operators, allowed the irradiation station to be successfully installed and safe methodologies to be defined for the installation and retrieval of the samples.

V. DOSIMETRY CALCULATIONS

FLUKA Monte Carlo software [36–38] simulations were used to estimate the total dose absorbed by the samples in the irradiation positions provided by the shelves. The scope of the simulations also included verification



FIG. 11. Mock-up trials of recovery scenarios: In the left picture Telemax is recovering some dropped containers; an intermediate support is on the floor with a sample having fallen out of it. Right side, simulation of three possible scenarios (A, B, C) of a dropped intermediate support. The yellow container was used to replicate the retention vessel in i-NEAR.

that the desired doses can be delivered with sufficient homogeneity and within time intervals compatible with the facility schedule.

A. Geometry and sample composition

A simplified geometry of the shelves, of the irradiation containers and of the samples has been implemented in FLUKA to simulate the radiation quantities of interest, as shown in Figure 12. In the simulation, the irradiation set-up is represented by a cylindrical container made of aluminium having outer diameter of 6.3 cm, inner diameter of 5.6 cm and height of 13.1 cm. The cylinder is generally filled with about 100 cm^3 of sample.

Table I shows the assumed sample composition employed for the Monte Carlo dose assessment. As described in Section VII, the irradiation station is used to irradiate a wide selection of commercial materials. As a first approximation and based on previous studies [23, 46], the selected composition can be considered as generally representative of polymeric materials mostly composed of light elements, such as lubricants and elastomeric O-rings. The chosen materials are mostly composed of carbon, hydrogen and oxygen, but metallic traces are usually present in uncontrolled amounts. As described in Section VI, these traces are included to simulate the residual activation of the samples after neutron irradiation, which is mainly due to metals.

B. Absorbed dose

In the twenty-four irradiation positions provided by the shelves, there is a mixed radiation field dominated by neutrons and photons. The main quantity selected to characterize the dosimetry of the irradiated positions is



FIG. 12. FLUKA Monte Carlo geometry of the area surrounding the n_TOF third generation spallation target, focusing on the shelves of the irradiation station.

TABLE I. Chemical composition of the polymeric material used for dose simulations.

Elem	ent	Mass $(\%)$
	С	83.9
	Η	10.0
	Ο	4.3
	Na	1.6
	Zn	0.177
	Ca	$3.64 \cdot 10^{-2}$
	Cd	$3.83 \cdot 10^{-4}$
	Br	$6.87 \cdot 10^{-5}$
	Cr	$2.95 \cdot 10^{-5}$
	La	$2.95 \cdot 10^{-5}$
	Ce	$2.95 \cdot 10^{-5}$
	$^{\mathrm{Sb}}$	$6.87 \cdot 10^{-6}$
	As	$4.91 \cdot 10^{-6}$
:	Sm	$3.93 \cdot 10^{-6}$
U-2	238	$3.90 \cdot 10^{-6}$
	\mathbf{Sc}	$1.96 \cdot 10^{-6}$
U-2	235	$2.83 \cdot 10^{-8}$

the physical absorbed dose in a reference sample along with its main contributions, the neutron dose and the gamma dose components. The gamma dose is computed as an electromagnetic dose which includes the contributions from photons, electrons and positrons. Since this contribution is in this case dominated by the gamma component, in the paper it will be simply referred to as gamma dose. All the other particles are neglected, as their contribution is several orders of magnitude lower.

Dose calculations are performed using a proton beam of energy 20 GeV/c [9, 52] impinging on the spallation target. It is convenient to express dose values in Gy/pulse, a nominal pulse corresponding to $7 \cdot 10^{12}$ protons [9, 53]. As a general reference based on previous experience, a total of approximately $2.5 \cdot 10^{19}$ protons on target (POT) are expected to be delivered in 200 working days during a normal physics production year, which corresponds to about 1 year of standard operation of the n_TOF facility.



FIG. 13. Total absorbed dose in the samples in the top shelf (a) and in the bottom shelf (b), averaged over each sample volume. Dose values are expressed in Gy/pulse, for a reference value of $2.5 \cdot 10^{19}$ protons on target (POT), expected to be delivered during a normal physics production year of 200 working days. The numbering of the samples corresponds to the different groups of 6 samples handled by the robots. The target, whose position is not shown in the Figure, is further down in x direction, meaning that samples 1-2-7-8 and 13-14-19-20 are the closest to the target (on the top shelf and bottom shelf, respectively).

Figure 13 shows the dose absorbed in the samples, averaged over the whole sample volume. On the top shelf, the total doses range from 0.5 Gy/pulse to 0.7 Gy/pulse, due to different distances from the target (x axis in the Figure). On the bottom shelf, the total doses range from 0.35 Gy/pulse to 0.45 Gy/pulse. In a standard physics operation year, the delivered doses correspondingly range from 1.2 MGy (lowest value in the bottom

shelf) to 2.5 MGy (highest value in the top shelf).

In both shelves, the samples located at the same distance from the target (such as samples in position 1, 2, 7 and 8 in Figure 13, and so on) absorb a comparable dose. By contrast, the total dose progressively decreases the farther the samples are laterally with respect to the target, as reported in Table II. In the Table, dose values are averaged over the four samples located at the same lateral distance from the target.

The gamma component of the dose is roughly independent of the specific irradiation position: it corresponds to 0.13 Gy/pulse in all the positions of the top shelf, while it corresponds to 0.11 Gy/pulse in all the positions of the bottom shelf. By contrast, the neutron component of the dose ranges from 67% to 81% of the total, progressively decreasing with distance from the target. This can be explained considering that organic materials act as moderators of high-energy neutrons, due to the high amount of hydrogen present in their composition. The samples located further from the target are partially shielded by the presence of the closer samples.

The dose gradients within the sample volumes are clearly visible in Figure 14, showing a mesh of total dose in the top shelf in different projections. The dose progressively decreases with the distance from the target along the x axis, while has higher homogeneity in the horizontal direction (z axis) and in the vertical one (y axis), as the distance from the target is roughly unchanged. Considering these gradients, an overall error of approx. $\pm 15\%$ on the total dose should be accounted for while averaging the dose on the whole sample volume.

The discussed calculations highlight that neutrondominated doses in the MGy range can be delivered in all the irradiation positions provided by the shelves, with a satisfactory homogeneity for macroscopic samples of up to 100 cm^3 of volume.

Since the neutron dose, and accordingly the ratio between neutron and gamma component, are highly dependent on the total amount of hydrogen, variations in respect to the reported values are expected when different materials are irradiated. Dedicated calculations using refined compositions of the irradiated materials will be performed for future specific irradiation campaigns.

C. Particle energy spectra

Figure 15 shows a typical neutron spectrum on one of the filled samples (number 4 of the top shelf) together with the proton, photon and electron spectra, whose fluences are several orders of magnitude lower compared to the neutron one. As can be seen, the neutron spectrum ranges from thermal up to about several GeV of maximum energy, and three main areas can be identified: thermal neutrons (~ 25 meV), evaporation (~ 1 MeV) and spallation (~ 100 MeV).

Other particles, whose contribution is orders of magnitude lower than the considered ones, such as muons and

TABLE II. Total dose in the samples, averaged over the four samples located at the same distance from the target, expressed in Gy/pulse. The neutron and gamma components of the dose are reported, as well as the ratio between the gamma component and the total. An overall uncertainty of about 15% is attributed to the dose values.

Shelf	Sample positions (average of four)	Total dose Gy/pulse	n dose Gy/pulse	γ dose Gy/pulse	$\begin{array}{c} \gamma \operatorname{comp.} \\ \% \end{array}$
Top	1, 2, 7, 8	0.68	0.55	0.13	19 %
Top	3, 4, 9, 10	0.54	0.41	0.13	$24 \ \%$
Top	5, 6, 11, 12	0.50	0.38	0.12	24 %
Bottom	13, 14, 19, 20	0.45	0.32	0.13	29 %
Bottom	15, 16, 21, 22	0.38	0.25	0.13	34 %
Bottom	17, 18, 23, 24	0.36	0.24	0.12	33 %



FIG. 14. Total dose in a selection of samples of the top shelf. a) Horizontal view of samples 1 to 12. b) Vertical view of samples 1, 2, 7, 8 (from left to right), having the same distance from the target. Dose values are reported in Gy/pulse.

pions, have been neglected.

D. Neutron fluence

Table III reports neutron fluence values calculated by FLUKA in correspondence of irradiation position number 4 on the top shelf, as a general reference. The total neutron fluence is reported, as well as its thermal and fast components. The fast neutron fluence accounts for neutrons with energy higher than 0.5 MeV. High energy neutrons are expected to be the most effective in delivering dose to organic hydrogenated materials, as the main energy transfer process is represented by elastic scattering of neutrons on hydrogen nuclei.

Thermal neutron fluence (n_{th} in Table III) is calculated following the standard used in electronics, accounting for all the neutrons with energies lower than 0.4 eV [54]. For completeness, thermal neutron equivalent fluence (n_{thEQ}) is reported in Table III as well. It is particularly meaningful in the radiation to electronics context, especially when the sensitive volume of a device contains ¹⁰B. This quantity is calculated by convoluting the differential neutron flux with a weighting function, whose value is one at 25 meV and decreases as the inverse of the square root of the energy [55, 56].

In comparison to the fast neutrons, the contribution of thermal neutrons to the total neutron dose deposited in organic hydrogenated materials is expected to be negligible. However, it could become more relevant in the case of different compositions of the irradiated materials, for example when the composition includes in relevant amounts atoms with a high capture cross-section for thermal neutrons, such as ${}^{10}B$ isotope.

VI. RADIATION PROTECTION CONSIDERATIONS

A. Radiological situation of the n_TOF NEAR and optimisation measures

Access to n_TOF NEAR is only possible when the facility is not operational and is subject to RP authorisation, following a dedicated radiological risk assessment. Residual dose rates at a-NEAR when the shielding is closed are typically in the order of 100-300 μ Sv/h after one day of



FIG. 15. Particle spectra in a reference position of the top shelf. For comparison with the dominating neutrons, the proton and electron spectra are multiplied by a factor 100. As a matter of comparison, the grey curve represents the neutron fluence in a-NEAR (multiplied by 10).

TABLE III. Neutron fluence per pulse calculated by FLUKA in irradiation position 4 (top shelf). The total neutron fluence is reported, as well as the thermal neutron equivalent fluence (ThnEq), the thermal fluence based on JEDEC standard[54] (ThN) and the fast neutron fluence.

n _{th}	$n_{\rm thEQ} \ (\rm JEDEC < 0.4 eV)$	$n_{\rm fast} \ (>0.5 \ {\rm MeV})$	Total
n/cm ² /pulse	$\rm n/cm^2/pulse$	$\rm n/cm^2/pulse$	n/cm ² /pulse
$2.8 \cdot 10^{10}$	$3.4 \cdot 10^{10}$	$1.0 \cdot 10^{10}$	$9.3 \cdot 10^{10}$

cool-down time and of 10-30 μ Sv/h after one week. However, access to the irradiation station requires the shielding to be open. The residual dose rate at i-NEAR can be in the order of several tens of mSv/h up to more than 100 mSv/h in close proximity to the spallation target. For this reason and according to the ALARA (As Low As Reasonably Achievable) principle, remote handling is deemed necessary to install and retrieve the samples, as described in Section IV. This contributes to significantly reducing the dose received by personnel, as the robotic equipment can be controlled from a distance in low-doserate areas (a few μ Sv/h).

B. Estimates of samples' induced radioactivity

In addition to the residual dose rate of the target area, the residual dose rate due to the induced activation of the samples is assessed; the results contribute to the definition of optimisation measures (such as additional cooldown time, shielding) in order to, first, safely retrieve and, then, handle the samples for post-irradiation analysis. Residual activation of samples and containers is estimated via FLUKA simulations. The following section provides an example of the RP studies performed for the first irradiation campaign started in 2021 (described in Section VII).

1. Simulation parameters

The FLUKA simulations for RP purposes were performed implementing the geometry of the n_TOF facility (as shown in Figure 12) and selecting the value of 7.6 10^{18} POT; this corresponds to the number of protons received by the n_TOF target during the first irradiation campaign organised in 2021, which lasted about four months (described in Section VII). However, the RP considerations reported here follow a conservative approach and can also be considered as generally representative of future irradiation campaigns.

The residual dose rate was simulated for a set of six samples installed on their intermediate support in the top shelf, where the highest activation is expected. The chemical composition described in Table I was used. As described in Section V, this composition, including an estimation of traces of metallic impurities, is generally representative of the organic materials to be irradiated in the shelves. Further considerations on the specific materials to be irradiated are reported in Section VII.

Different products or even the same product from a different batch generally have slightly different compositions and concentrations of metallic traces and therefore it is impossible to precisely predict their amount and the resulting activation.

2. Residual dose rate

Figure 16 shows the residual dose rate as a function of the cool-down time (from 1 day up to 10 years) obtained from the FLUKA Monte Carlo simulations. The residual dose rate at 40 cm from the intermediate support containing a set of six containers and their samples reduces from about 500 μ Sv/h after one day down to 10 μ Sv/h after one week. This is related to the decay of Na-24 (half-life of about 15 hours), which is produced in the aluminium intermediate support, in the irradiated containers and in the samples (from Na traces). Based on these results and according to the ALARA principle, a minimum cool-down time of one week is implemented prior to the removal of the samples. A comparison between the simulated values and the results measured after the first irradiation is given in Section VII.



FIG. 16. 1-D distribution along the FLUKA z-axis of the residual dose rate for various cool-down times (from 1 day up to 10 years).

3. Radionuclide inventory and radioactive waste aspects

To evaluate the radionuclide inventory of a sample, the particle spectral fluences scored at the location of the samples were folded with nuclide production cross sections of the lubricant material by means of the ActiWiz analytical code [57]. By providing ActiWiz with the specific irradiation profile, the radionuclide inventory is obtained for the cool-down time of interest. The sample radio-toxicity is analysed in terms of the Swiss clearance limit (*Limite de Libération* in French or LL), and the licensing limit, (*Limite d'Autorisation* in French or LA. The inhalation of activity at the licensing limit on a single occasion yields a committed effective dose of 5 mSv. [58]. The LA indicates the value corresponding to the absolute activity level of a material above which handling of this material is subject to mandatory licensing and shall be performed in laboratories (called "Work Sectors") complying with specific requirements in terms of, e.g., radiation and fire protection, and ventilation.

Table IV provides the fraction of LL and LA as a function of the cool-down time together with the main contributors, which are Na-22, Na-24 (only for a cool-down time of 1 day) and Zn-65 (originating from impurities of Na and Zn present in the samples). Even after ten years of cool-down time the fraction of LL for a sample is higher than one, which means that it cannot be cleared from regulatory control and it should therefore be disposed of as a radioactive waste. However, it should be noted that the final conclusion strongly depends on the amount of impurities; it cannot be excluded that very pure samples could become conventional waste after irradiation. This will be evaluated after the sample removal, combining the results from the FLUKA simulations with dedicated γ -spectrometry measurements.

After 1 week of cool-down the total activity in the sample is already less than 1 LA and therefore sample handling inside a laboratory classified as work sector according to ref. [58] would not be required.

VII. FIRST IRRADIATION OF MATERIALS

A. Pilot irradiation completed in 2021

The irradiation station was successfully installed at $n_{-}TOF$ in June 2021, and the first samples were installed for irradiation in time for the first operation of the new spallation target, in July 2021. Figure 10 shows Telemax installing the first set of samples, in Figure 5 the four sets of samples installed in the NEAR Irradiation station before irradiation are shown. The irradiated materials represent a selection of commercial lubricants and elastomeric materials of interest for various applications at CERN. Details on the selected products and on the post-irradiation characterization of the samples will be described in dedicated publications.

The first pilot irradiation allowed the technical and safety aspects of the developed methodology to be verified. Measurements of the residual activation of the containers and of the different irradiated materials were performed after irradiation. After about four months of irra-

Nuclide	$t_{1/2}$	t_c	t_c	t_c	t_c	t_c	t_c
	7	1 d	1 week	1 month	1 y	3у	10 y
LL							
Na-22	2.6 y	$1.07 \cdot 10^{3}$	$1.07 \cdot 10^{3}$	$1.05 \cdot 10^{3}$	$8.42 \cdot 10^2$	$4.84 \cdot 10^2$	$7.52 \cdot 10$
		$<\!1\%$	2%	2%	4%	17%	92%
Na-24	15 h	$3.89 \cdot 10^5$	$4.82 \cdot 10^2$	< 0.1	< 0.1	< 0.1	< 0.1
		87%	$<\!\!1\%$	$<\!\!1\%$	$<\!1\%$	$<\!1\%$	$<\!1\%$
Zn-65	$244~{\rm d}$	$5.32 \cdot 10^4$	$5.23 \cdot 10^4$	$4.90 \cdot 10^4$	$1.89 \cdot 10^4$	$2.83 \cdot 10^{3}$	1.68
		12%	95%	96%	95%	82%	2%
Total		$4.45 \cdot 10^5$	$5.50 \cdot 10^4$	$5.09 \cdot 10^4$	$1.98 \cdot 10^4$	$2.89 \cdot 10^{3}$	$8.21 \cdot 10$
LA							
Na-24	15 h	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1
		94%	2%	$<\!1\%$	$<\!1\%$	$<\!1\%$	$<\!1\%$
Zn-65	$244~{\rm d}$	0.3	0.3	0.2	0.1	$<\!1\%$	$<\!1\%$
		6%	90%	93%	95%	82%	$<\!1\%$
Total		4.6	0.3	0.3	0.1	$<\!1\%$	$<\!1\%$

TABLE IV. Multiple of the clearance limit (limite de libération (LL), top) and of licensing limit (limite d'autorisation (LA), bottom) for the main radionuclides produced in the lubricant sample as a function of the cool-down time t_c The relative contribution to the total LL is given as well.

diation (less than the nominal duration of 200 days due to 2021 being a target commissioning year, hence of shorter duration), according to the simulations discussed in Section V, samples absorbed doses approximately ranging between 0.4 MGy and 0.8 MGy. These doses refer to a generic material composition which, based on previous experimental analyses, is considered to be roughly representative of organic materials. The dose values here reported are intended to be general estimations only. Dedicated simulations taking into account the specific composition and irradiation position of each sample will be performed and the results will be detailed in future works, along with the post-irradiation analyses.

The first sets of samples were successfully removed from the irradiation facility using Telemax, confirming the safe design of the irradiation facility and the feasibility of the whole procedure.

B. Residual activity after irradiation

The measured residual dose rates of the intermediate support and of its respective six containers and samples after the first irradiation confirm that the containers can be safely and easily handled by human operators after some weeks of cooling time.

The residual dose rate of the most-activated intermediate support, measured after removal of the sample containers and after 23 days of cooling time was of 320 μ Sv/h at contact, of 35 μ Sv/h at 10 cm of distance and of 8 μ Sv/h at 40 cm of distance. The residual dose rate of the most radioactive containers, including inner container and corresponding sample, were measured. Depending on the specific distance from target, the overall residual dose rate for single containers irradiated in the top shelf after 23 days of cooling time ranged between 70 μ Sv/h and 110 μ Sv/h at contact, between 7 μ Sv/h and 10 μ Sv/h at a distance of 10 cm and of a few μ Sv/h at a distance of 40 cm. The order of magnitude of these measured values matches the simulated values discussed in Section VI.

These values fully confirm the possibility of safely following the whole life cycle of the irradiated samples and containers in agreement with RP principles.

C. Container opening

After irradiation, samples were examined under a fume hood. Custom-designed reachers (as shown in Figure 8) are used to extract the inner container from the outer one, allowing a distance of 40 cm to be maintained between the operator and the container, if needed. The reachers also enable extraction of the inner container without spilling non solid samples. The whole handling system is designed to be managed remotely. However, after the first irradiation the dose rate was sufficiently low to allow both the inner and the outer containers to be directly handled by human operators.

Grease and of oil samples were examined after irradiation. The grease samples maintained their position in the inner containers, as confirmed by the absence of spills and by the absence of contamination in the inner containers lids. A small fraction of the oil sample (some ml) spilled out of the inner container, however it was successfully contained by the outer container, thus avoiding any contamination of the environment.

D. Dose measurements: readout of RPL dosimeters

As mentioned in Section IIIC, eight RPL dosimeters have been irradiated during the pilot irradiation completed in 2021. Figure 6 shows the position of the dosimeters on the irradiation station, while Figure 17 reports their readout measurement in comparison to the total dose simulated by FLUKA in the dosimeters themselves.



FIG. 17. The figure shows the ratio between the simulated dose absorbed by the dosimeters and the experimental dose measurement, showing an overall good agreement within the experimental uncertainties.

The doses measured by the dosimeters range from 100 kGy to 150 kGy, showing a satisfactory agreement with the simulated absorbed dose in the dosimeters, correspondingly ranging from 135 kGy to 200 kGy. The simulated doses systematically overestimate the measured ones by about 50% on average, however, the measured and simulated values are overall compatible within the errors.

The dosimeters' readouts are roughly independent of the specific irradiation position within the irradiation station. By contrast, large differences are reported in the simulations of the dose absorbed by samples of organic materials in different positions, as discussed in Section V. This can be explained considering the different composition of the dosimeters and of the samples, and the correspondingly different ratio between gamma and neutron dose.

In the dosimeters, the neutron component of the dose represents on average 10% of the total absorbed dose only, while in the organic samples it represents about 70-80%. These differences are motivated by the much higher content of hydrogen and other light elements in organic samples, which highly influence the dose delivered by neutrons, due to moderation processes. This component is greatly reduced in the dosimeters, since they are made of phosphate glass and their composition is hydrogen free and carbon free. The gamma component of the dose, by contrast, is less dependent on these differences in the composition. In this scenario, the homogeneity of the dosimeter readouts can be explained. The dose gradients observed in the simulated samples, as reported in Table II, are in fact mostly due to the differences in the neutron component of the dose.

This considered, the dose measured by the dosimeters can roughly be compared to the gamma component of the dose absorbed by the samples during the pilot irradiation, corresponding to about 140 kGy in all the positions, achieving a very satisfactory agreement.

The dose measured by the dosimeters can accordingly be considered as a good estimate of the gamma dose absorbed by organic materials in the irradiation station.

E. Current and future irradiation studies on a selection of commercial materials

The samples chosen for the first irradiation represent a selection of commercial products currently being studied and used at CERN, including both radiation tolerant and generic ones.

Currently, studies performed in the framework of the Radiation to Materials (R2M) activities focus on lubricants (in the form of greases and of fluid oils) and on elastomeric materials presently available in the market. In fact, they represent critical components necessarily used in high radiation areas [10], and whose failure can have a high impact on accelerator operation [8]. In particular, polyphenyl ether (PPE)-based lubricants are being irradiated and studied in detail, as their composition is known to be promising for radiation tolerance [1, 20, 59]. Recent studies have confirmed the superior resistance of PPE-based products [46] and have evidenced possible differences between gamma and reactor mixed field in determining radiation damage [23]. The data collected at n_TOF NEAR in the first pilot irradiation and in future irradiation campaigns will contribute in further understanding radiation damage mechanisms.

VIII. CONCLUSIONS

Construction and installation of a new CERN multipurpose irradiation station was successfully completed during July 2021 in the n_TOF NEAR area. The new irradiation station takes advantage of the spallation neutrons produced by the recently installed third-generation n_TOF spallation target and has minimal impact on the facility operation and physics programme.

As a result, CERN is now equipped with several irradiation positions for materials, where unprecedented radiation damage data can be collected in a neutrondominated environment. The available irradiation conditions are - for the first time - much closer, in comparison with standard gamma irradiations, to the ones encountered during operation in several accelerator facilities worldwide.

The data produced thanks to this new station will allow a better assessment of the radiation tolerance of commercial and custom-made materials, so far largely lacking or undocumented, and permit a corresponding increase of the lifetime of devices used in high-radiation areas, thereby reducing the risk of failures during operation of accelerators and other facilities with similar challenges.

Radiation safety is a key aspect in the design and operation of the irradiation station and post-irradiation analysis of samples. Simulations, including FLUKA Monte Carlo techniques, were extensively used to predict dosimetry in the samples and the resulting induced activation. Remote handling via telemanipulated robots was integrated into the design and operating procedures at an early stage and extensive mock-up testing was used to demonstrate the safety of the whole sample life cycle. Sample containers and their supports were subjected to in-depth risk analysis and practical testing before obtaining approval to carry out the first pilot irradiation.

Successful pilot irradiation of material samples over a

four month period and subsequent inspections and measurements on the irradiated samples demonstrated the effectiveness of the design of the irradiation station and its sample containers along with the safe and reliable methodology for sample irradiation and handling.

ACKNOWLEDGMENTS

The authors would like to acknowledge the support of CERN's Sources, Targets and Interactions (STI) Group, the Radiation-to-Electronics (R2E) Project as well as the Accelerator Consolidation (ACC-CONS) Project at CERN. The authors would like to express thanks to S. Sorlut for the design of the NEAR collimator as well as to D. Boon for the construction of the target mobile shielding.

- R. O. Bolt and J. G. Carrol. Radiation Effects on Organic Materials. Academic Press, 1963.
- [2] Tadao Seguchi and Yosuke Morita. Radiation Resistance of Plastics and Elastomers. Takasaki Radiation Chemistry Research Establishment, VI:583–589, 2003. Japan Atomic Energy Research Institute JAERI, Gunma 370-12 Japan. URL: https://doi.org/10.1002/ 0471532053.bra047.
- [3] Makiko Saito, Katsunori Anzai, Takahito Maruyama, Yuto Noguchi, Kenichi Ueno, Nobukazu Takeda, and Satoshi Kakudate. Development of radiation hard components for ITER blanket remote handling system. Fusion Engineering and Design, 109-111:1502–1506, 2016. URL: https://doi.org/10.1016/j.fusengdes. 2015.11.042.
- [4] Matteo Ferrari, Aldo Zenoni, Yong Joong Lee, and Alberto Andrighetto. Neutron Radiation Effects on Lubricants and O-rings for Target and Accelerator Applications. Spallation Materials Technology, Materials Science Forum, 1024:127–133, 2021. URL: https://doi.org/10. 4028/www.scientific.net/MSF.1024.127.
- [5] R Catherall, W Andreazza, M Breitenfeldt, A Dorsival, G J Focker, T P Gharsa, Giles T J, J-L Grenard, F Locci, P Martins, S Marzari, J Schipper, A Shornikov, and T Stora. The ISOLDE facility. *Journal of Physics G: Nuclear and Particle Physics*, 44(9):094002, aug 2017. doi:10.1088/1361-6471/aa7eba.
- [6] Lucia Popescu, Donald Houngbo, and Marc Dierckx. High-power target development for the next-generation ISOL facilities. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 463:262–268, 2020. URL: https: //doi.org/10.1016/j.nimb.2019.05.023.
- [7] Duchemin, Charlotte and Ramos, Joao P. and Stora, Thierry and Ahmed, Essraa and Aubert, Elodie and Audouin, Nadia and Barbero, Ermanno and Barozier, Vincent and Bernardes, Ana-Paula and Bertreix, Philippe and Boscher, Aurore and Bruchertseifer, Frank and Catherall, Richard and Chevallay, Eric and Christodoulou, Pinelopi and Chrysalidis, Katerina and Cocolios, Thomas E. and Comte, Jeremie and Crepieux,

Bernard and Deschamps, Matthieu and Dockx, Kristof and Dorsival, Alexandre and Fedosseev, Valentin N. and Fernier, Pascal and Formento-Cavaier, Robert and El Idrissi, Safouane and Ivanov, Peter and Gadelshin, Vadim M. and Gilardoni, Simone and Grenard, Jean-Louis and Haddad, Ferid and Heinke, Reinhard and Juif, Benjamin and Khalid, Umair and Khan, Moazam and Köster, Ulli and Lambert, Laura and Lilli, G. and Lunghi, Giacomo and Marsh, Bruce A. and Palenzuela, Yisel Martinez and Martins, Renata and Marzari, Stefano and Menaa, Nabil and Michel, Nathalie and Munos, Maxime and Pozzi, Fabio and Riccardi, Francesco and Riegert, Julien and Riggaz, Nicolas and Rinchet, Jean-Yves and Rothe, Sebastian and Russell, Ben and Saury, Christelle and Schneider, Thomas and Stegemann, Simon and Talip, Zeynep and Theis, Christian and Thiboud, Julien and van der Meulen, Nicholas P. and van Stenis, Miranda and Vincke, Heinz and Vollaire, Joachim and Vuong, Nhat-Tan and Webster, Benjamin and Wendt, Klaus and Wilkins, Shane G. and , the CERN-MEDICIS collaboration. CERN-MEDICIS: A Review Since Commissioning in 2017. Frontiers in Medicine, 8, 2021. URL: https://www.frontiersin.org/article/ 10.3389/fmed.2021.693682, doi:10.3389/fmed.2021. 693682

- [8] Matteo Ferrari, Dominika Senajova, Keith Kershaw, Antonio Perillo Marcone, and Marco Calviani. Selection of radiation tolerant commercial greases for high-radiation areas at CERN: methodology and applications. *Journal* of Nuclear Materials and Energy, 29:101088, 2021. URL: https://doi.org/10.1016/j.nme.2021.101088.
- [9] R. Esposito, M. Calviani, O. Aberle, M. Barbagallo, D. Cano-Ott, T. Coiffet, N. Colonna, C. Domingo-Pardo, F. Dragoni, R. Franqueira Ximenes, L. Giordanino, D. Grenier, F. Gunsing, K. Kershaw, R. Logé, V. Maire, P. Moyret, A. Perez Fontenla, A. Perillo-Marcone, F. Pozzi, S. Sgobba, M. Timmins, and V. Vlachoudis (for the n_TOF Collaboration). Design of the thirdgeneration lead-based neutron spallation target for the neutron time-of-flight facility at CERN. *Physical Review Accelerators and beams*, 24:093001, 2021. URL: https:

//doi.org/10.1103/PhysRevAccelBeams.24.093001.

- [10] J. Maestre, C. Torregrosa, K. Kershaw, C. Bracco, T. Coiffet, M. Ferrari, R. Franqueira Ximenes, S. Gilardoni, D. Grenier, A. Lechner, V. Maire, J.M. Martin Ruiz, E. Matheson, N. Solieri, A. Perillo-Marcone, T. Polzin, V. Rizzoglio, D. Senajova, C. Sharp, M. Timmins, and M. Calviani. Design and behaviour of the Large Hadron Collider external beam dumps capable of receiving 539 MJ/dump. Journal of Instrumentation, 16:P11019, 2021. URL: https://iopscience.iop.org/ article/10.1088/1748-0221/16/11/P11019.
- [11] R. Catherall, W. Andreazza, M. Breitenfeldt, A. Dorsival, G.J. Focker, T.P. Gharsa, T.J. Giles, J.-L. Grenard, F. Locci, P. Martins, S. Marzari, J. Schipper, A. Shornikov, and T. Stora. The ISOLDE facility. Journal of Physics G: Nuclear and Particle Physics, 44:094002, 2017. URL: https://iopscience.iop.org/ article/10.1088/1361-6471/aa7eba.
- [12] Tim Giles. Isolde V. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 463:254-257, 2020. URL: https: //doi.org/10.1016/j.nimb.2019.05.025.
- [13] Obara Kenjiro, Kakudate Satoshi, and Oka Kiyoshi. High Gamma-Rays Irradiation Tests of Critical Components for ITER (International Thermonuclear Experimental Reactor) In-vessel Remote Handling System. Japan Atomic Energy Research Inst., Tokyo (Japan), 1999. Technical report - JAERI-Tech-99-003. URL: https: //www.osti.gov/etdeweb/servlets/purl/354703.
- [14] M.B. Bruce and M.V. Davis. Radiation effects on Organic Materials in Nuclear Plants. 1981. Georgia Institute of Technology, Nuclear Engineering Department EPRI NP-2129, Research Project 1707-3, Final Report. URL: https://doi.org/10.2172/5591289.
- [15] W.L.R. Rice and W.L. Cox. The Effects of Nuclear Radiation in Solid Film Lubricants. 1959. Materials Laboratories of the Wright Air Development Center, Air Research and Development Command, United State Air Force, Ohio, WADC Technical Report 58-499 ASTIA Document No. 207795. URL: https://www.osti.gov/ biblio/4195204.
- [16] H. Shulman and W.S. Ginell. Nuclear and Space Radiation Effects on Materials. 1970. NASA Space Vehicle Design Criteria, NASA SP-8053. URL: https: //ntrs.nasa.gov/citations/19710015558.
- [17] S.B. Srivastava and V.G. Thomas. Lubricants in nuclear technology: a brief review. 1974. Government of India, Atomic Energy Commission, Bhabha Atomic research Centre, Bombay, India. URL: https://inis.iaea.org/ search/search.aspx?orig_q=RN:6212859.
- [18] A. Schönbacher H.and Stolarz-Izycka. Compilation of Radiation Damage Test Data: Cable Insulating Materials. 1979. European Organisation for Nuclear Research (CERN) Geneva, Switzerland, Yellow Report CERN-79-04, Part 1. URL: https://cds.cern.ch/record/ 133188/files/CERN-HS-RP-038-YR-PARTI.pdf.
- [19] A. Schönbacher H.and Stolarz-Izycka. Compilation of Radiation Damage Test Data: Thermosetting and thermoplastic resins. 1979. European Organisation for Nuclear Research (CERN) Geneva, Switzerland, Yellow Report CERN-79-08, Part 2. URL: https://cds.cern.ch/record/120566/files/ CERN-HS-RP-038-YR-PARTII.pdf.

- [20] Beynel P., Maier P., and Schönbacher H. Compilation of Radiation Damage Test Data: Materials used around high-energy accelerators. 1982. European Organisation for Nuclear Research (CERN) Geneva, Switzerland, Yellow Report CERN-82-10, Part 3. URL: https://cds. cern.ch/record/141784/files/CERN-HS-RP-093.pdf.
- [21] B.A. Briskman. Specificity of proton irradiation effects on polymers. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 265, Issue 1:72-75, 2007. URL: https://doi.org/10.1016/j.nimb.2007.08.028.
- [22] Agnès Rivaton and Jack Arnold. Structural modifications of polymers under the impact of fast neutrons. *Polymer Degradation and Stability*, 93, Issue 10:1864–1868, 2008. URL: https://doi.org/10.1016/ j.polymdegradstab.2008.07.015.
- [23] Matteo Ferrari, Aldo Zenoni, Yongjoong Lee, and Yoshikazu Hayashi. Characterization of a polyphenyl ether oil irradiated at high doses in a TRIGA Mark II nuclear reactor. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 497:1-9, 2021. URL: https://doi. org/10.1016/j.nimb.2021.03.021.
- [24] IAEA Consultancy Meeting on 'Radiation effects on polymer materials'. 2019. IAEA International Atomic Energy Agency, EVT1905348, 28-29 November 2019, Vienna, Meeting report. URL: http://www-naweb. iaea.org/napc/iachem/working_materials/IAEA% 20Consultancy%20Meeting%20Final%20Report.pdf.
- [25] Rubén García Alía, Markus Brugger, Francesco Cerutti, Salvatore Danzeca, Alfredo Ferrari, Simone Gilardoni, Yacine Kadi, Maria Kastriotou, Anton Lechner, Corinna Martinella, Oliver Stein, Yves Thurel, Andrea Tsinganis, and Slawosz Uznanski. Lhc and hl-lhc: Present and future radiation environment in the high-luminosity collision points and rha implications. *IEEE Transactions on Nuclear Science*, 65(1):448–456, 2018. doi: 10.1109/TNS.2017.2776107.
- [26] Fiona Harden, Aymeric Bouvard, Nikolaos Charitonidis, Yacine Kadi (on behalf of the HiRadMat experiments, and facility support teams). Targetry Challenges & HiRadMat. JPS Conference Proceedings, 33:011149, 2021. Proceedings of the 3rd J-PARC Symposium (J-PARC2019). URL: https://doi.org/10.7566/JPSCP. 33.011149.
- [27] Federico Ravotti, Blerina Gkotse, Maurice Glaser, Pierre Jouvelot, Isidre Mateu, Viktoria Meskova, Giuseppe Pezzullo, and Jean-Michel Sallese. The IRRAD Proton Irradiation Facility Control, Data Management and Beam Diagnostic Systems: An Outlook of the Major Upgrades Beyond the CERN Long Shutdown 2. In Proc. ICALEPCS 2019, New York, USA, Oct. 5-11, 2019, ISBN 978-3-95450-209-7, 2019. URL: https://accelconf.web.cern.ch/icalepcs2019/ doi/JACoW-ICALEPCS2019-WEPHA127.html.
- [28] Adam Thornton. CHARM Facility Test Area Radiation Field Description. 2016. European Organisation for Nuclear Research (CERN) Geneva, Switzerland, ATS Department, CERN-ACC-NOTE-2016-12345, 2016-04-28. URL: https://cds.cern.ch/record/ 2149417/files/CHARM%20Facility%20Test%20Area% 20Radiation%20Field%20Description.pdf.
- [29] n_TOF CERN website. Https://ntof-exp.web.cern. ch/.

- [30] C. Borcea, P. Cennini, M. Dahlfors, A. Ferrari, G. Garcia-Munoz, P. Haefner, A. Herrera-Martinez, Y. Kadi, V. Lacoste, E. Radermacher, F. Saldana, V. Vlachoudis, L. Zanini, C. Rubbia, S. Buono, V. Dangendorf, R. Nolte, and M. Weierganz. Results from the commissioning of the n_tof spallation neutron source at cern. Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, 513(3):524– 537, 2003. URL: https://www.sciencedirect.com/ science/article/pii/S0168900203020722, doi:https: //doi.org/10.1016/S0168-9002(03)02072-2.
- [31] Raffaele Esposito and Marco Calviani. Design of the third-generation neutron spallation target for the CERN's n_TOF facility. Journal of Neutron Research, 22, no. 2-3:221-231, 2020. URL: https://content.iospress.com/articles/ journal-of-neutron-research/jnr190137.
- [32] n_TOF Collaboration. CERN n_TOF Facility: Performance Report, INTC-2002-037. Jan 2003. URL: https: //cds.cern.ch/record/1258424.
- [33] The n_TOF Collaboration. The new n_TOF NEAR Station. CERN-INTC-2020-073 / INTC-I-222, 2020. Letter of Intent to the ISOLDE and Neutron Timeof-Flight Committee, European Organisation for Nuclear Research (CERN) Geneva, Switzerland, CERN-INTC-2020-073 INTC-I-222. URL: https://cds.cern. ch/record/2737308/files/INTC-I-222.pdf.
- [34] M. Cecchetto, M. Sacristan Barbero, G. Lerner, R. Garcia Alia, D. Senajova, F. Garcia Infantes, J.A. Pavon Rodriguez, M. Sabate Gilarte, and V. Vlachoudis. Electronics Irradiation with Neutrons at the NEAR Station of the n_TOF Spallation Source at CERN. accepted for RADECS 2022, European Conference on Radiation and its Effects on Components and Systems, 2022.
- [35] Carlo Rubbia, Samuel A Andriamonje, D Bouvet-Bensimon, S Buono, R Cappi, P Cennini, C Gelès, I Goulas, Y Kadi, P Pavlopoulos, Jean Pierre Charles Revol, A Tzima, and V Vlachoudis. A High Resolution Spallation Driven Facility at the CERN-PS to Measure Neutron Cross Sections in the Interval from 1 eV to 250 MeV: a Relative Performance Assessment. Technical report, CERN, Geneva, Jun 1998. Addendum to CERN-LHC-98-002-EET. URL: https://cds.cern.ch/ record/363828.
- [36] Giuseppe Battistoni, Till Boehlen, Francesco Cerutti, Pik Wai Chin, Luigi Salvatore Esposito, Alberto Fassò, Alfredo Ferrari, Anton Lechner, Anton Empl, Andrea Mairani, Alessio Mereghetti, Pablo Garcia Ortega, Johannes Ranft, Stefan Roesler, Paola R. Sala, Vasilis Vlachoudis, and George Smirnov. Overview of the FLUKA code", journal = "Annals of Nuclear Energy. 82:10–18, 2015. Joint International Conference on Supercomputing in Nuclear Applications and Monte Carlo 2013, SNA + MC 2013. URL: https://doi.org/10.1016/j.anucene. 2014.11.007.
- [37] T.T. Böhlen, F. Cerutti, M.P.W. Chin, A. Fassò, A. Ferrari, P.G. Ortega, A. Mairani, P.R. Sala, G. Smirnov, and V. Vlachoudis. The FLUKA Code: Developments and Challenges for High Energy and Medical Applications. *Nuclear Data Sheets*, 120:211–214, 2014. doi: https://doi.org/10.1016/j.nds.2014.07.049.
- [38] FLUKA.CERN website. https://fluka.cern/.

- [39] Julien Ordan. Installation of material samples at the new n_TOF NEAR irradiation station. Jul 2021. General Photo. URL: https://cds.cern.ch/record/2774594.
- [40] H. Vincke, I. Brunner, I. Floret, D. Forkel-Wirth, M. Fuerstner, S. Mayer, and C. Theis. Response of alanine and radio-photo-luminescence dosemeters to mixed high-energy radiation fields. *Radiation Protection Dosimetry*, 125, Issue 1-4:340–344, 2007. doi:https: //doi.org/10.1093/rpd/ncm157.
- [41] A. Brynjolfsson and N.W. Holm. The Co 60 Irradiation Facility and the Gamma Field at Risø. In *Large radiation* sources in industry, 1960.
- [42] J. Trummer. Characterisation of CERN's High Level Dosimetry Radio-Photo-Luminescence readout system. Technical Note, EDMS: 2275199, CERN, 2020.
- [43] Oak Ridge National Laboratory ORNL website, In-Vessel Irradiation, High Flux Isotope Reactor HFIR. https:// neutrons.ornl.gov/suites/in-vessel-irradiation.
- [44] A. Zenoni, F. Bignotti, A. Donzella, G. Donzella, M. Ferrari, S. Pandini, A. Andrighetto, M. Ballan, S. Corradetti, M. Manzolaro, A. Monetti, M. Rossignoli, D. Scarpa, D. Alloni, M. Prata, A. Salvini, and F. Zelaschi. Radiation resistance of elastomeric O-rings in mixed neutron and gamma fields: Testing methodology and experimental results. *Review of Scientific In*struments, 88:113304, 2017. doi:https://doi.org/10. 1063/1.5011035.
- [45] OTTO FLUID CONTROL GMBH website. https:// www.otto-fluid-control.de/.
- [46] Matteo Ferrari, Aldo Zenoni, Monika Hartl, Yongjoong Lee, Alberto Andrighetto, Alberto Monetti, Andrea Salvini, and Fabio Zelaschi. Experimental study of consistency degradation of different greases in mixed neutron and gamma radiation. *Heliyon*, 5, Issue 9:E02489, 2019. doi:https://doi.org/10.1016/j.heliyon.2019. e02489.
- [47] Teodor eod webpage. https://www.telerob.com/en/ products/teodor-evo.
- [48] Telemax evo pro webpage. https://www.telerob.com/ en/products/telemax-family/telemax-evo-pro/page.
- [49] Di Castro, Mario and Buonocore, Luca Rosario and Ferre, Manuel and Gilardoni, Simone and Losito, Roberto and Lunghi, Giacomo and Masi, Alessandro. A dual arms robotic platform control for navigation, inspection and telemanipulation. page TUPHA127. 5 p, 2018. URL: https://cds.cern.ch/record/2305665, doi:10.18429/JACoW-ICALEPCS2017-TUPHA127.
- [50] Vanden Broeck, Renilde. ROBOTIC SOLUTIONS -CERNBot Robotic Base (Robots Pool). Sep 2019. URL: https://cds.cern.ch/record/2695197.
- [51] Telerob website. https://www.telerob.com/en/.
- [52] C. Guerrero et al. Performance of the neutron time-offlight facility n_TOF at CERN. The European Physical Journal A, 49:27, 2013. doi:https://doi.org/10.1140/ epja/i2013-13027-6.
- [53] M. Barbagallo et al. High-accuracy determination of the neutron flux at n_tof. The European Physical Journal A, 49:156, 2013. doi:https://doi.org/10.1140/epja/ i2013-13156-x.
- [54] JEDEC Solid State Technology Association. Measurement and reporting of alpha particle and terrestrial cosmic ray-induced soft errors in semiconductor devices. *JEDEC Standard, JESD89A*, Oct 2006, Riaffirmed Jan 2012. URL: https://www.jedec.org/

standards-documents/docs/jesd-89a.

- [55] M. Cecchetto, R. Garcia Alia, S. Gerardin, M. Brugger, A. Infantino, and S. Danzeca. Impact of thermal and intermediate energy neutrons on sram see rates in the lhc accelerator. *IEEE Trans. Nucl. Sci.*, 65, 8:1800–1806, Aug 2018. URL: https://ieeexplore. ieee.org/document/8352861.
- [56] R. G. Alía, M. Brugger, S. Danzeca, J. Mekki, and A. Thornton. See cross section calibration and application to quasi-monoenergetic and spallation facilities. in Proc. EPJ Web Conf, 153:08015, Sept 2017. URL: https://www.epj-conferences.org/ articles/epjconf/abs/2017/22/epjconf_icrs2017_ 08015/epjconf_icrs2017_08015.html.
- [57] Vincke, H. and Theis, C. ActiWiz 3 an overview of the latest developments and their application. J.

Phys.: Conf. Ser., 1046:012007. 10 p, 2018. URL: https://cds.cern.ch/record/2636327, doi:10.1088/1742-6596/1046/1/012007.

- [58] Ordonnance sur la radioprotection (ORaP) du 26 avril 2017. https://www.fedlex.admin.ch/eli/cc/2017/ 502/fr.
- [59] Kazuo Arakawa and Tsuneo Sasuga and Miyuki Hagiwara and Naohiro Hayakawa and Kenzo Yoshida and Hiroshi Nakanishi and Mitsuo Hirohama and Tetsuya Yagi and Tamio Akada and Takao Soda. Super highly radiation-resistant grease. United States Patent,, Patent Number: US-4753741-A:012007. 10 p, jun. 28, 1988. URL: https://patents.google.com/ patent/US4753741A.